Author's response to reviews

Title: Patterns of active and passive smoking, and associated factors in the southeast Anatolian project (SEAP) region, in Turkey

Authors:

Ali Ihsan Bozkurt (abozkurt@pamukkale.edu.tr)
Saime Sahinoz (saimo2000@hotmail.com)
Birgul Ozciirci (ozciirci@gantep.edu.tr)
Servet Ozgur (ozgur@gantep.edu.tr)
Turgut Sahinoz (turgut29@mynet.com)
Hamit Acemoglu (hamitacemoglu@dicle.edu.tr)
Gunay Saka (gsaka@dicle.edu.tr)
Ali Ceylan (alic@dicle.edu.tr)
Yilmaz Palanci (yilmazp@dicle.edu.tr)
Ersen Ilcine (ilcine@dicle.edu.tr)
Feridun Akkafa (feridun@harran.edu.tr)

Version: 4  Date: 23 November 2005

Author's response to reviews: see over
Dear Editor,

Please find attached a revised paper named “Patterns Of Active and Passive Smoking And Associated Factors in South-east Anatolian Project (SEAP) Region, Turkey”. We revised each comment and provided answers as follows.

Thank you for the opportunity to submit our work. Yours truly,

Ali Ihsan Bozkurt, MD, Ph.D.

The Department of Public Health, Faculty of Medicine, Pamukkale University,
Kınıklı, Denizli, TURKEY
e-mail: abozkurt@pamukkale.edu.tr
Tel: +90 258 2138177-2138178,
Fax: +90 258 2132874
Reviewer

General

1. The manuscript highly improved after the II revision. I only suggest authors to delete the following two sentences, that, although correct, appear to be in contrast with the conclusions of the Abstract:

   a) End of 6th paragraph of Discussion: “This suggests that anti smoking campaigns in SEAP region should target women especially”.

   b) End of Conclusions: "Young people and women seem to be appropriate target for anti-smoking campaigns."

Answer: We deleted these two sentences.